Author's response to reviews

Title: Heterotopic Ossification after Patellar Tendon Repair presenting in Trisomy 8 Mosaicism: a case report and literature review

Authors:

    Austin W Chen (achen9@uic.edu)
    Samuel J Chmell (schmell@uic.com)

Version: 5 Date: 28 July 2011

Author's response to reviews: see over
Date: 7/5/2011

To the editors of the Journal of Medical Case Reports:

*Clarify whether, given that patients presenting with Trisomy 8 Mosaicism can sometimes be mentally retarded and have developmental difficulties, the patient in this case report was able to fully comprehend the significance of providing consent to publish. If not, please ask the authors to seek consent to publish from the patient's next of kin.

The patient’s mother was the party that provided consent.

*Blank out all writing (except for any annotations relevant to the case report) from the figures (e.g. in the corners).

This has been done.

Regards,

~Austin Chen

Austin Chen, MD
University of Illinois at Chicago
Department of Orthopaedic Surgery